Answer for Editor
April 14, 2022

Title: The mechanism for development of malnutrition in primary biliary cholangitis and the principles of its correction

Authors: Vasily Ivanovich Reshetnyak, Igor Veniaminovich Maev

Name of Journal: World Journal of Meta-Analysis

ESPS Manuscript NO: 75746

Dear Jin-Lei Wang, Company Editor-in-Chief, Editorial Office,

Thank you very much for the editing of the manuscript.

My ID 00068912. The authors send you revised the manuscript 75746 "The mechanism for development of malnutrition in primary biliary cholangitis and the principles of its correction" (Please refer to the attached file “75746_revised manuscript”)

All your comments and requirements were taken into account.

The authors express their gratitude to the reviewers (ID – 03537165; ID – 03538156; for their work done and comments.

Response to the reviewer ID – 03537165.
The authors express their gratitude to the reviewer for the work done and for the positive evaluation of the manuscript. Methods to assess the status of nutrition, this is a separate big topic for discussion. Since the presented manuscript is devoted to the mechanism of development of malnutrition in patients with primary biliary cholangitis, the authors did not focus on methods for assessing the nutritional status. We really hope that the reviewer will agree with this argument.

Response to the reviewer ID – 03538156.
Once again, the authors express their gratitude for the detailed analysis of the manuscript and the comments made aimed at improving the manuscript.

Thank you for publishing our manuscript in the World Journal of Meta-Analysis.

Sincerely yours,

Vasiliy Ivanovich RESHETNYAK, professor, DSc, MD;
Department of propaedeutic of internal diseases and gastroenterology, A.I. Yevdokimov Moscow State University of Medicine and Dentistry. Delegatskaya St., 20, p. 1, 127473, Moscow, Russian Federation. vasiliv.reshetnyak@yandex.ru

Telephone: +7-495-6096700
Fax: +7-495-6812229